\chapter{Summary of Commands}
\label{summary_of_commands}

\begin{tabular}{r c l}
    {\tt version}     & Page \pageref{version_cmd}     & Prints the currently running version of MPX-OS. \\
    {\tt date}        & Page \pageref{date_cmd}        & Prints or updates the system date. \\
    {\tt directory}   & Page \pageref{directory_cmd}   & Prints a listing of MPX programs that may be run. \\
    {\tt stop}        & Page \pageref{stop_cmd}        & Stops MPX-OS and returns to DOS. \\
    {\tt help}        & Page \pageref{help_cmd}        & Prints help on MPX commands. \\
    {\tt prompt}      & Page \pageref{prompt_cmd}      & Changes the MPX prompt. \\
    {\tt alias}       & Page \pageref{alias_cmd}       & Adds an alias for an MPX command. \\
    {\tt show}        & Page \pageref{show_cmd}        & Displays a table of MPX processes. \\
    {\tt load}        & Page \pageref{load_cmd}        & Loads a program. \\
    {\tt resume}      & Page \pageref{resume_cmd}      & Resumes a suspended program. \\
    {\tt run}         & Page \pageref{run_cmd}         & Runs a program. \\
    {\tt suspend}     & Page \pageref{suspend_cmd}     & Puts a program into the suspended state. \\
    {\tt terminate}   & Page \pageref{terminate_cmd}   & Terminates a program, freeing its resources. \\
    {\tt setpriority} & Page \pageref{setpriority_cmd} & Sets the execution priority for a program. \\
    {\tt clock}       & Page \pageref{clock_cmd}       & Prints or updates the system time. \\
\end{tabular}
